Author's response to reviews

Title: The impact of nurses' working hours on patient safety culture - A Cross National Survey among Japan, the United States, and Chinese Taiwan using Hospital Survey on Patient Safety Culture-

Authors:

Yinghui Wu (wuyinghui@med.toho-u.ac.jp)
Shigeru Fujita (sfujita@med.toho-u.ac.jp)
Kanako Seto (setokana@med.toho-u.ac.jp)
Shinya Ito (s1007m09@med.toho-u.ac.jp)
Kunichika Matsumoto (rakchart@med.toho-u.ac.jp)
Chiu-Chin Huang (cchuang@must.edu.tw)
Tomonori Hasegawa (tommie@med.toho-u.ac.jp)

Version: 2 Date: 19 March 2013

Author's response to reviews: see over
2013.3. 19

Dear Dr. Natalie Pafitis,
Executive editor,
BMC Health Services Research

I wrote this letter to you as I made a change in my manuscript that an “Ethical concerns” was added in the method section base on the requirement from BioMed Central Editorial. Please check the revised manuscript and many thanks for you in advanced.

I am looking forward to hearing from you.

Yours sincerely

(On behalf of the authors)
Yinghui WU
Department of Social Medicine,
Toho University School of Medicine.
5-21-16 Omori-nishi, Ota-ku,
Tokyo 143-8540, Japan
Tel: +81-3-3762-4151 ext.2415
Fax: +81-3-5493-5417
e-mail: wuyinghui@med.toho-u.ac.jp